Author’s response to reviews

Title: "I Always Tried to Forget about the Condition and Pretend I was Healed": Coping with Cervical Cancer in Rural Ghana

Authors:

Charity Binka (cbinka@gmail.com)
Samuel Nyarko (samharrenson@gmail.com)
Kofi Awusabo-Asare (k.awusabo-asare@ucc.edu.gh)
David Doku (dokudavid@gmail.com)

Version: 4 Date: 23 Jan 2018

Author’s response to reviews:

Author’s response to editor

Title: “I Always Tried to Forget about the Condition and Pretend I was Healed”: Coping with Cervical Cancer in Rural Ghana.

Authors: Charity Binka, Samuel Harrenson Nyarko, Kofi Awusabo-Asare, David Teye Doku

Version: 3

Date: 23 January 2018

Dear Editor,

Thank you for the opportunity to revise our manuscript based on your comments. We have carefully considered your suggestions and have consequently revised the manuscript accordingly. Please see below the point-by-point responses to the suggestions.

With kind regards,

Charity Bink, PhD
Corresponding author.

Author's response to editor’s comments

1. Please have the text edited by a professional language editing service or a native English speaking colleague. There are many issues with grammar, wording, spelling, and/or punctuation that need to be addressed before acceptance for publication.

  We have checked the entire manuscript for grammatical and typographical errors. We let an English language expert proofread the manuscript. We believe the manuscript has improved and look forward to your kind acceptance for publication.

2. Please consider the list of authors as it currently stands with reference to our guidelines regarding qualification for authorship. Currently, the contributions of authors SHN, KAA, and DTD are not clear and do not automatically qualify them for authorship. Please provide clarification on their contributions, or remove their names from the list of authors and place them in the “Acknowledgements” section instead.

  Thank you very much for your correction. We can confirm that the previously stated contributions of the authors were not correctly stated. We have, therefore, revised it to make the contributions of the authors clear as indicated below. (See authors’ contributions in the manuscript). The following is the corrected version:

  CB, KAA, and DTD conceived and designed the study. CB led the data collection with technical support from KAA and DTD. CB and SHN performed the analysis with technical support from KAA and DTD. CB and SHN led the initial draft of the manuscript with technical support from KAA and DTD. All authors were involved in the critical review of the manuscript for important intellectual content. All authors read and approved the final version of the manuscript for publication.